Author's response to reviews

Title: Effects of epidural lidocaine analgesia on labor and delivery: a randomized, prospective, controlled trial

Authors:
Shahram - Nafisi - (nafisish@yahoo.com)

Version: 5 Date: 11 December 2006

Author's response to reviews: see over
Dear Annabel Phillips

Thanks from your email and the comments of professor Halpern.

All revisions, based on the comments of professor Halpern was performed in the text.

Sincerely,
Shahram Nafisi MD